Author’s response to reviews

Title: Case Report: Mechanisms of HIV Elite Control in two African Women

Authors:

Yumna Moosa (yumbilina@gmail.com)
Ramla Tanko (tanko.fatimeramla@uct.ac.za)
Veron Ramsuran (veron.ramsuran@gmail.com)
Ravesh Singh (Singhra@ukzn.ac.za)
Nonhlanhla Yende-Zuma (Nonhlanhla.Yende@caprisa.org)
Melissa-Rose Abrahams (melissa-rose.abrahams@uct.ac.za)
Philippe Selhorst (Philippe.Selhorst@uct.ac.za)
Kamini Gounder (gounderk@ukzn.ac.za)
Carolyn Williamson (carolyn.williamson@uct.ac.za)
Salim Abdool Karim (Salim.AbdoolKarim@caprisa.org)
Nigel Garrett (Nigel.Garrett@caprisa.org)
Wendy Burgers (wendy.burgers@uct.ac.za)

Version: 1 Date: 04 Sep 2017

Author’s response to reviews:

INFD-D-17-00955 (BMC Infectious Diseases)

Case Report: Mechanisms of HIV Elite Control in two African Women

Yumna Moosa; Ramla F Tanko; Veron Ramsuran; Ravesh Singh; Nonhlanhla Yende-Zuma; Melissa-Rose Abrahams; Philippe Selhorst; Kamini Gounder; Carolyn Williamson; Salim S Abdool Karim; Nigel J Garrett; Wendy A Burgers
Dear Philippa

We have now obtained Consent to Publish from both participants in our manuscript, and have stated this in the notes section. We have also made one other change, and that is to rename the two participants from the original PID numbers, to “Elite controller 1” and “Elite controller 2”, as requested by our IRB.

The revised manuscript has been uploaded.

warmest wishes

Wendy Burgers

Corresponding Author